\section{GENERAL CORE-DESCRIPTION MODULES}\label{sect:modesc1}

\input{SectRESINI}
\vskip 1.0cm
\input{SectUSPLIT}
\vskip 1.0cm
\input{SectMACINI}
\vskip 1.0cm
\input{SectDEVINI}
\vskip 1.0cm
\input{SectDETINI}
\vskip 1.0cm
\input{SectLZC}
\vskip 1.0cm
\input{SectDSET}
\vskip 1.0cm
\input{SectMCC}
\vskip 1.0cm
\input{SectMOVDEV}
\vskip 1.0cm
\input{SectNEWMAC}
\vskip 1.0cm
\input{SectFLPOW}
\vskip 1.0cm
\input{SectTAVG}
\vskip 1.0cm
\input{SectTINST}
\vskip 1.0cm
\input{SectSIM}
\vskip 1.0cm
\input{SectXENON}
\vskip 1.0cm
\input{SectDETECT}
\vskip 1.0cm
\input{SectCVR}
\vskip 1.0cm
\input{SectHST}
\vskip 1.0cm
\input{SectROD}
\vskip 1.0cm
\input{SectIDET}

\section{CROSS-SECTION INTERPOLATION MODULES}\label{sect:modesc2}

\input{SectCRE}
\vskip 1.0cm
\input{SectNCR}
\vskip 1.0cm
\input{SectSCR}
\vskip 1.0cm
\input{SectACR}
\vskip 1.0cm
\input{SectMCR}
\vskip 1.0cm
\input{SectPCR}
\vskip 1.0cm
\input{SectAFM}
\vskip 1.0cm
\input{SectT16CPO}
\vskip 1.0cm
\input{SectD2P}
\vskip 1.0cm

\section{THERMAL-HYDRAULICS MODULES}\label{sect:modesc3}

\input{SectTHM}

\section{POINT KINETICS MODULES}\label{sect:modesc4}

\input{SectPKINI}
\input{SectPKINS}

\section{OPTIMIZATION MODULES}\label{sect:modesc5}

This section is related to optimization capabilities available in Donjon and
based on generalized perturbation theory.\cite{optex1,optex2} General information
about the generalized perturbation theory can be found in Sect. 5.3 of Ref.~\citen{PIP2016}.

\input{SectDLEAK}
\vskip 1.0cm
\input{SectDSPH}
\vskip 1.0cm
\input{SectDREF}
\vskip 1.0cm
\input{SectGRAD}
\vskip 1.0cm
\input{SectPLQ}
\vskip 1.0cm
\input{SectLNSR}

\section{PIN-POWER RECONSTRUCTION MODULES}\label{sect:modesc6}

This section is related to pin-power reconstruction capabilities available in Donjon.
The corresponding theory is explained in \cite{Chambon2014,Fliscounakis2011} 

\input{SectNAP}

